ŗ



# **Geophysical Research Letters**

### **RESEARCH LETTER**

10.1029/2019GL084730

#### Key Points:

- Groundwater accounts for a large fraction of streamflow and only receives minor (~2%) recharge from glaciers in the study catchment in Peru
- As meltwater decreases, groundwater provides consistent discharge in the near term (~30 years), becoming a larger fraction of streamflow
- In the long term (>60 years), groundwater storage and discharge decrease in response to higher evapotranspiration and lower precipitation

#### **Supporting Information:**

- Supporting Information Figure S1
- Figure S1
- Figure S2
- Figure S3
- Figure S4

#### Correspondence to:

L. D. Somers, lsomers@mit.edu laurendsomers@gmail.com

#### Citation:

Somers, L. D., McKenzie, J. M., Mark, B. G., Lagos, P., Ng, G.-H. C., Wickert, A. D., et al. (2019). Groundwater buffers decreasing glacier melt in an Andean watershed—but not forever. *Geophysical Research Letters*, 46, 13,016–13,026. https://doi.org/10.1029/ 2019GL084730

Received 26 JUL 2019 Accepted 27 OCT 2019 Accepted article online 9 NOV 2019 Published online 19 NOV 2019

©2019. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Groundwater Buffers Decreasing Glacier Melt in an Andean Watershed—But Not Forever

Lauren D. Somers<sup>1,2</sup>, Jeffrey M. McKenzie<sup>1</sup>, Bryan G. Mark<sup>3</sup>, Pablo Lagos<sup>4</sup>, Gene-Hua Crystal Ng<sup>5,6</sup>, Andrew D. Wickert<sup>5,6</sup>, Christian Yarleque<sup>4</sup>, Michel Baraër<sup>7</sup>, and Yamina Silva<sup>4</sup>

<sup>1</sup>Department of Earth and Planetary Sciences, McGill University, Montreal, Quebec, Canada, <sup>2</sup>Now at the Department of Civil and Environmental Engineering, Massachusetts Institute of Technology, Cambridge, MA, USA, <sup>3</sup>Byrd Polar and Climate Research Centre, Ohio State University, Columbus, OH, USA, <sup>4</sup>Instituto Geofísico del Perú, Lima, Peru, <sup>5</sup>Department of Earth Sciences, University of Minnesota, Minneapolis, MN, USA, <sup>6</sup>Saint Anthony Falls Laboratory, University of Minnesota, Minneapolis, MN, USA, <sup>7</sup>Département de Génie de la Construction, École de Technologie Supérieure, Montreal, Quebec, Canada

**Abstract** Accelerating mountain glacier recession in a warming climate threatens the sustainability of mountain water resources. The extent to which groundwater will provide resilience to these water resources is unknown, in part due to a lack of data and poorly understood interactions between groundwater and surface water. Here we address this knowledge gap by linking climate, glaciers, surface water, and groundwater into an integrated model of the Shullcas Watershed, Peru, in the tropical Andes, the region experiencing the most rapid mountain-glacier retreat on Earth. For a range of climate scenarios, our model projects that glaciers will disappear by 2100. The loss of glacial meltwater will be buffered by relatively consistent groundwater discharge, which only receives minor recharge (~2%) from glacier melt. However, increasing temperature and associated evapotranspiration, alongside potential decreases in precipitation, will decrease groundwater recharge and streamflow, particularly for the RCP 8.5 emission scenario.

**Plain Language Summary** Mountain regions play an important role in water supply, because meltwater from snow and ice feeds rivers during dry periods. Groundwater (water stored in the pore spaces of soils and rock), which flows into rivers, is also an important store of water in mountain areas and may help to protect water resources against the negative impacts of shrinking mountain glaciers. We used extensive field measurements and computer modeling of the Shullcas Watershed in the Peruvian Andes to determine the current and future role of groundwater in the face of climate change. Our model projects that glaciers in our study area will disappear by 2100. The loss of glacier meltwater is buffered in the short term (~30 years) by consistent groundwater flow to rivers. However, in the long term (>60 years), precipitation is expected to decrease and rising temperatures lead to increased evaporation and water use by plants. These factors reduce groundwater recharge and storage, causing dry season streamflow to drop.

### **1. Introduction**

Mountains sustain water resources by gradually and consistently supplying snow and ice melt for downstream communities. However, accelerating glacier recession in a warming climate threatens the sustainability of these natural water towers (Barnett et al., 2005; Viviroli et al., 2007).

Glacier retreat has accelerated worldwide since the 1980s (Kaser et al., 2006; Paul et al., 2004), with tropical alpine glaciers experiencing the most rapid areal recession (Bradley et al., 2006; Paul et al., 2004; Rabatel et al., 2013; Vuille et al., 2018). The Peruvian Andes, which host 70% of the world's tropical glaciers, have had coverage decrease by over 40% since the 1970s (Autoridad Nacional del Agua, 2014). Communities and industries in the Andes and on the arid Pacific coast are vulnerable to climate-driven intensification of dry season (May–September) water shortages (Vuille et al., 2018), when only 20% of annual precipitation occurs in the Peruvian Andes (Bury et al., 2013).

Groundwater, previously often ignored in mountain hydrology (Bales et al., 2006; Huss et al., 2008; Juen et al., 2007; Ragettli et al., 2014; Voeckler et al., 2014), is now understood to be a significant contributor to streamflow generation in the Andes (Baraer et al., 2015; Glas et al., 2018; Gordon et al., 2015; Saberi et al., 2019; Somers et al., 2016) and in mountain systems globally (Andermann et al., 2012; Frisbee et al., 2011;

## **Geophysical Research Letters**



**Figure 1.** Conceptual model of an Andean proglacial watershed showing hydrologic pathways and future changes in water resources. A combination of glacier melt and groundwater discharge currently generates dry season streamflow. In the future, air temperature will continue to increase, while changes in precipitation are more uncertain. In response, glacier melt will decrease and eventually disappear, ET will increase, and groundwater storage will decrease (i.e., lower water table), all leading to lower stream discharge for all modeled climate scenarios.

Hood & Hayashi, 2015; Liu et al., 2004; McClymont et al., 2010). While groundwater supplies may prove more resilient than snow and glaciers to climate change (Lemieux et al., 2008; Tague et al., 2008), our knowledge of future impacts is limited by oversimplification of groundwater processes in hydrological models due computational and/or observational limitations in mountain regions (Juen et al., 2007; Ragettli et al., 2014). While there have been some implementations of integrated surface-subsurface hydrologic models in glacierized watersheds (Lemieux et al., 2008; Saberi et al., 2019), it is unclear how results from these few studies generalize to other settings or extrapolate into the future as glaciers retreat.

Here we directly address this research gap by developing a modeling approach that integrates distributed groundwater flow and recharge with surface-water and glacier-melt processes. We constrain our model with extensive field observations from 1998 to 2018. Using our integrated modeling approach, we first elucidate the role of mountain groundwater in the proglacial Andes by characterizing dominant streamflow contributors and flow paths, as well as quantifying glacier-groundwater interactions. Secondly, we apply climate projections to the integrated model to assess the capacity of groundwater to buffer the hydrological impacts of climate change and mountain glacier recession.

#### 2. Study Area

The Shullcas Watershed (area =  $170 \text{ km}^2$ ), central Peru, is a typical proglacial watershed (Figure 1), and is comprised of glacierized peaks, steep alpine grasslands, and flatter valley-bottom wetlands. It is underlain by glacio-fluvial sediments above metamorphic rock of the Huaytapallana Complex in the north and weakly metamorphosed sedimentary rock of the Mitu, Cabanilla, and Pucara Groups in the south (Chew et al., 2016). The steep alpine grasslands are dominated by *Jarva ichu* grass, while the flat alpine wetlands (known as bofedales) host cushion-forming vegetation such as *Disticha muscoides* and other plants including *Plantaeog rigida*, and *Jarva ichu* (Maldonado Fonkén, 2014; Salvador et al., 2014). The Shullcas River receives meltwater from the glaciers of the Cordillera Huaytapallana (Figure 2a), which have retreated by 44% between 1984 and 2011 (López-Moreno et al., 2014). The Shullcas River is the primary water source for the city of Huancayo (population: 365 K; Instituto Nacional de Estadística y Informática, 2012), which



**Figure 2.** Maps of the Shullcas Watershed. (a, inset) Location of the city of Huancayo (population: 365,000), which relies on the nearby, proglacial Shullcas Watershed for the majority of its water supply. (a) Land surface elevation, locations of field measurements, and Huaytapallana Glaciers in the north. (b) Simulated water-table elevation for 1 January 2000 mimics the land surface elevation. (c) GSFLOW simulation results of change in water-table elevation between average levels for 2000–2009 and 2090–2099 inclusive, using Eta regional climate model projections downscaled from the MIROC5 GCM RCP 4.5.

experiences frequent water shortages during the dry season (Mark et al., 2017). The social and economic importance of this watershed and availability of climate, glacier, hydrologic, and hydrogeological data make the Shullcas Watershed an ideal test site to use a system approach to assess the impact of climate change on Andean hydrology.

#### 3. Methods

#### 3.1. Data Collection and Processing

We obtained daily maximum and minimum air temperature and precipitation, from the National Meteorology and Hydrology Service of Peru (SENAMHI) for two meteorological stations (Huaytapallana, 4,684 m above sea level, and Shullcas, 3,839 m above sea level; Figure 2a) within the watershed from 2008 to 2018 (SENAMHI, 2018). Data from two additional weather stations (Huayao and Santa Ana) located within 20 km of the watershed and from NASA's Tropical Rainfall Measurement Mission (TRMM version 3B42 RR; Huffman & Bolvin, 2015) were used to extend temperature and precipitation inputs, respectively, back to 1998 and fill data gaps using linear regression to the primary stations.

We obtained monthly streamflow data at the watershed outflow (Location 6; Figure 2a) from 1985 to 2009 from the Peruvian National Water Authority (ANA). In June 2015, we installed five stream-discharge gauges and two lake-level gauges. In July 2016, we installed an additional stream gauging station and seven shallow (<2.5 m) groundwater wells in two clusters (Figure 2a).

#### 3.2. Integrated Hydrological Model Setup

We integrated GSFLOW v2.0.0 (Markstrom et al., 2008), a coupled groundwater–surface-water model, with a glacier melt module to simulate the complete hydrological system of the Shullcas Watershed. GSFLOW combines MODFLOW (Langevin et al., 2017), a finite-difference groundwater flow model, and PRMS (Markstrom et al., 2015), a distributed-parameter, process-based, surface water model. GSFLOW has a daily time step, and we prepared our GSFLOW input files using GRASS-GSFLOW version 1.0.0 (Ng et al., 2018). Input parameters were based on field measurements where possible, and literature values otherwise (see supporting information and Table S1). For our purposes, we define the shallow unsaturated zone as the top ~0.5 m of the subsurface, which is controlled by PRMS within GSFLOW. The MODFLOW domain above the water table is defined as the deep unsaturated zone and the MODFLOW domain below the water table is the saturated zone.

We discretized the surface water component of the model into 45 hydrologic response units (HRUs) classified as either glacier/rock, grassland, or grassland/wetland based on a supervised maximum likelihood classification of Landsat satellite imagery (after Mather, 1985). We used the ASTER Global Digital Elevation Model with 30-m resolution (Tachikawa et al., 2011) to define the watershed boundaries and topography. Precipitation and temperature were spatially distributed based on elevation lapse rates calculated from the two meteorological stations within the watershed. Given the lack of humidity and wind speed data, potential evapotranspiration (PET) was calculated using a Priestley-Taylor approach (Liu et al., 2017; Priestley & Taylor, 1972). Actual evapotranspiration (ET) was calculated in GSFLOW according to how much water was available in the capillary reservoir of the soil zone. After ET, excess water in the soil zone could go to interflow, based on fast and slow interflow coefficients, and to the deeper subsurface MODFLOW domain (See Markstrom et al., 2008).

The subsurface was discretized horizontally into 2,850, 242 by 246-m grid cells. We impose an upper layer thickness of 20 m based on road cut observations that exposed over 10 m of poorly sorted soils. Beneath this, the fractured bedrock layer extends an additional 180 m. Below 200-m depth, we assume that bedrock is impermeable. We placed three constant-head MODFLOW cells downstream of the watershed outlet point to allow for some groundwater outflow from the modeled domain and prevent forced groundwater discharge to the stream network.

We used a semidistributed, modified temperature-index model to simulate glacier melt and mass balance for the Huaytapallana glaciers (after Quick & Pipes, 1977; see supporting information S3 for more details). Temperature and precipitation were calculated for each 100-m elevation band per time step. Melt, m, for each elevation band, b, was calculated as

$$m_b = \begin{cases} MF_s(T - T_{crit}), T > T_{crit} \\ 0, T \le T_{crit} \end{cases}$$
(1)

where  $MF_s$  is the melt factor of a given season *s* (dry or rainy), *T* is the average daily temperature (°C), and  $T_{crit}$  is the minimum temperature where melt is possible (-1 °C in this case). We calibrated seasonal melt factors to match glacial lake outflow for Chuspicocha Lake from 2015 to 2018. At each time step, *t*, the mass balance of the glacier, *Mass*, is calculated as

$$Mass_t = Mass_{t-1} + \rho \sum_{1}^{b} ((Sn_b - m_b - sub_b)^* A_b)$$
<sup>(2)</sup>

where is the density of water (1,000 kg/m<sup>3</sup>), Sn is the total daily snowfall (m water equivalent), Sub is the daily sublimation (m water equivalent), and A is the area of the elevation band (m<sup>2</sup>).

We ran the integrated model over two time periods. From 1998 to 2018, we drove the model with meteorological observations from within the Shullcas Watershed for model calibration and analysis of hydrograph components. During this period, we also ran the model with and without glacier melt to determine glacier melt contribution to groundwater discharge. From 1961 to 2099, we drove the model using downscaled GCM outputs for hydrologic projections.

#### 3.3. Model Calibration and Sensitivity Analysis

We calibrated selected input parameters (using >100 model runs) to minimize an error metric (equation (3)) that combines multiple data sources to ensure that the model adequately captures relevant processes across the modeled domain. Twenty key parameters were varied within a range of possible values (Table S1). The root-mean-square error (RMSE) was calculated for each available streamflow record and included in an aggregate error,  $E_{Agg}$ :

$$E_{Agg} = \frac{E_{1998-2009} + E_1 + E_2 + \dots + E_6}{7}$$
(3)

where  $E_{1998-2009}$  is the RMSE of the monthly discharge record from ANA from 1998–2009 and  $E_{1-6}$  are the calculated RMSE for each of the six daily discharge stations between 2015 and 2018.

We used the Nash-Sutcliffe Efficiency  $(E_{N-S})$  to express model fit, where

$$E_{N-S} = \frac{\sum_{i=1}^{n} (Q_i^{obs} - Q_i^{sim})^2}{\sum_{i=1}^{n} (Q_i^{obs} - Q^{obs})^2}$$
(4)

 $Q^{obs}$  and  $Q^{sim}$  are observed and simulated stream discharge (m<sup>3</sup>/s), respectively, at any time step *i*.

Lastly, we used sensitivity analysis to establish error bounds based on four parameters that we identified as relatively uncertain given our inability to measure them across the modeled domain: (1) vertical and horizontal hydraulic conductivity of bedrock, (2) Priestly-Taylor  $\alpha$  coefficients for PET, (3) interflow coefficients, and (4) glacier melt factors.

Given the computational intensity of the model, we use a computationally efficient, one-at-a-time sensitivity analysis (Zheng & Bennett, 2002) to assess the uncertainty of the results (see supporting information).

#### 3.4. Climate Projections

Climate projections were generated from the dynamic Eta regional climate model (Chou, Lyra, Mourão, Dereczynski, Pilotto, Gomes, Bustamante, Tavares, Silva, Rodrigues, Campos, Chagas, Sueiro, Siqueira, Nobre, et al., 2014, Chou, Lyra, Mourão, Dereczynski, Pilotto, Gomes, Bustamante, Tavares, Silva, Rodrigues, Campos, Chagas, Sueiro, Siqueira, & Marengo, 2014) by the Center for Weather Forecasting and Climate Studies (CPTEC) at the Brazilian National Institute for Space Research (INPE; http://projeta. cptec.inpe.br). The Eta model downscales the MIROC5 (Model for Interdisciplinary Research on Climate), CanESM2 (Canada Earth System Model), and BESM (Brasilian Earth System Model) GCMs under historical and future (RCP4.5 and RCP8.5) scenarios with a spatial resolution of 20 km at a daily time step from 1961 to 2099. The BESM projections for RCP 4.5 were generated discontinuously and are therefore not used in our study.

Daily temperatures were corrected using the additive (classical) bias correction technique (Bordoy & Burlando, 2013; Hawkins et al., 2012) while daily precipitation raw data were corrected in two steps. First, linear scaling (Smitha et al., 2018), also called multiplicative shift (Ines & Hansen, 2006), bias correction were used, with a three-month sliding window precipitation. Second, the quantile mapping bias correction technique (Ines & Hansen, 2006) was applied following the procedure by Grillakis et al. (2013). The bias-corrected RCM results produce comparable glacier melt amounts to the local meteorological data.

#### 4. Results and Discussion

#### 4.1. Model Calibration and Sensitivity Analysis

Our calibrated, integrated hydrological model adequately represents field observations where  $E_{N-S}$  for stream discharge ranged from -0.46 to 0.74 (Figure S1 and Table S2), with four of the six gauging stations having  $E_{N-S}$  greater than 0.5. A hydrological model is often considered to adequately represent observed data if  $E_{N-S} > 0.5$  (Moriasi et al., 2007). Furthermore, temporally averaged modeled water-table depth closely matches observations from our two groundwater well clusters with root-mean-square errors (RMSE) of 0.20 and 1.50 m for the Chuspicocha and Huacracocha areas, respectively, from 2016 to 2018. The model underestimates the observed range in water table fluctuation, in part due to the large grid cells which may not accurately represent the point locations of the wells which are all located in wetlands close to the river for accessibility. Future work should seek to expand groundwater head data, particularly in highlands where piezometer and well data are extremely scarce.

Dry and rainy season glacier melt factors of 2.8 and 3.0 mm d<sup>-1</sup> °C<sup>-1</sup>, respectively, yielded the best fit to observed meltwater production. The relatively low melt factors are consistent with the dry, high-radiation setting where more of the available energy is partitioned to sublimation (Fernàndez & Mark, 2016; Hock, 2003).

The climatic setting of the tropical Andes lends itself well to the calibration of a joint surface-water and groundwater model. Because very little precipitation falls during the dry season, the baseflow recession pattern and dry-season streamflow serve as ready calibration targets for the groundwater-flow parameters. Meanwhile, the rainy season hydrograph serves as a good calibration target for the interflow parameters.





**Figure 3.** Modeled stream discharge contributors and storage reservoirs over a hydrologic year in the Shullcas Watershed. (a) Stacked plot of streamflow generation pathways. Groundwater discharge to the stream comes from the saturated zone, interflow to the stream comes from the shallow unsaturated zone. Groundwater recharge passes from the shallow unsaturated zone, through the deep unsaturated zone, to the saturated zone. (b) Cumulative change in water storage for modeled reservoirs. Unsaturated storage has approximately net zero storage change while the saturated zone has a negative net storage change, indicating that it is dominated by decadal trends in recharge.

Sensitivity analysis indicates that model results are most sensitive to the Priestly-Taylor  $\alpha$  coefficient, followed by the hydraulic conductivity of the fractured bedrock layer, the interflow coefficient, and the glacier melt factors (Table S3). Simulated streamflow has an uncertainty of ±23% for a 95% confidence interval (see supporting information).

#### 4.2. Groundwater Dominates Dry Season Streamflow

From 1998 to 2018, simulated historical groundwater discharge from the saturated zone to the stream contributes an average of 37% (ranging from -4.4 to 93%) of outlet streamflow annually and 72% during the dry season (Figure 3a). During the rainy season, interflow (shallow, preferential flow through the unsaturated zone) dominates (60%) streamflow. These results complement work by others (e.g., Andermann et al., 2012; Baraer et al., 2015; Hood & Hayashi, 2015; Pohl et al., 2015; Saberi et al., 2019; Somers et al., 2016) to demonstrate that groundwater is an important contributor to streamflow in glacierized mountain catchments.

Conversely, our calibrated model simulations indicate that glacier melt contributes only 8.0% of Shullcas River outflow discharge from 1998 to 2018. To validate this result, we compare the model outputs against hydro-chemical mixing analysis for streamflows from 10–22 July 2014 (Crumley, 2015). Our simulated 11% glacier-melt contribution to stream discharge for this specific time period lies within the geochemically estimated 8.9–17% streamflow contribution.

#### 4.3. Glacier Melt Contributes Little to Groundwater Recharge

Only a small amount (2.3%) of saturated zone groundwater discharge to the Shullcas River was originally sourced from glaciers that covered 2.4% of the watershed area in 2015. The low contribution of glacier melt to groundwater suggests that overall, groundwater recharge is resilient to a reduction in glacier meltwater input. To our knowledge, the only other quantitative estimate of glacier melt contribution to groundwater comes from a headwater catchment on Volcán Chimborazo in Ecuador, where 15% of groundwater discharge was found to originate from glaciers that cover 34% of the watershed area (Saberi et al., 2019). These results suggest a useful relationship between glacier coverage as a proportion of watershed area, and glacier-sourced groundwater discharge, similar to the relationship proposed by Baraer et al. (2015) between glacierized area and glacier melt contribution to streamflow. This relationship likely varies with climate, bedrock lithology, and topography (Markovich et al., 2016; Saberi et al., 2019; Tague et al., 2008), and requires more estimates in different basins to expound.

The simulated groundwater table (Figure 2b) remains close to the ground surface in valley bottoms and near streams, but is located as deep as 176 m below the ground surface in highland and steep areas (similar spatial trends to Ofterdinger et al., 2014), well below the 20-m-thick unconsolidated surficial aquifer. Therefore, much of the modeled groundwater flow toward the streams passes through the bedrock aquifer, as suggested in other mountain hydrological studies (Andermann et al., 2012; Liu et al., 2004; Tague et al., 2008). The longer and slower flow paths through bedrock allow groundwater to contribute to streamflow year round.

# 4.4. Unsaturated Zone Storage Fluctuates Seasonally While Saturated Groundwater Storage Follows Decadal Trends

Subsurface water storage fluctuates seasonally, increasing during the rainy season and decreasing during the dry season. For example, in 2016–2017 the range in cumulative annual storage change is largest in the deep unsaturated zone (0.0158 km<sup>3</sup>), followed by the shallow unsaturated zone (0.0095 km<sup>3</sup>) and the saturated zone (0.0060 km<sup>3</sup>; Figure 3b). For comparison, the nearby city of Huancayo requires approximately 0.023 km<sup>3</sup> of water per year, based on the 175 L per person-day estimated by the United Nations (United Nations Development Report, 2006).

The timing of changes in storage is important for water resource management. The unsaturated zone is quickly depleted after the end of the rainy season, whereas the saturated zone continues to drain and

supply water to streams throughout the dry season, during maximum water stress. The negative net change in saturated storage in a particular year (Figure 3b) illustrates the decadal time scale on which groundwater fluctuates and therefore its ability to buffer interannual climatic variability (Cuthbert et al., 2019), even in a mountain environment where hydraulic gradients are high.

#### 4.5. Rapid Projected Climate Warming Causes Disappearance of Huaytapallana Glaciers

Observed temperatures at the Huayao Meteorological Station from 1975 to 2005 increased at a rate of 0.42 °C per decade. The averaged projected rates of warming from the Eta RCM over the twenty-first century are 0.46 and 0.88 °C per decade for RCP 4.5 and 8.5 emission scenarios, respectively (Figure 4a). The high projected rates of warming are consistent with elevation-dependent warming (Mountain Research Initiative EDW Working Group, 2015; Seth et al., 2010; Urrutia & Vuille, 2009; Yarleque et al., 2018). For the RCP 4.5 scenario, minimum and maximum temperatures begin to stabilize around 2070. For RCP 4.5, little change is projected in precipitation and the GCMs disagree on the direction of change, whereas RCP 8.5 projects more consistent negative trends in precipitation.

All emission scenarios and GCMs run through our glacier melt module project the complete disappearance of the Huaytapallana glaciers by the end of the twenty-first century (Figure 4a), with the slowest melt scenario projecting disappearance (glacier area <100 m<sup>2</sup>) in 2085. Our glacier projections are comparable to equilibrium-line-elevation projections for the Quelccaya ice cap (Yarleque et al., 2018), approximately 400 km southeast of our study area, and with a similar summit elevation (5,680 versus 5,557 m for the Huaytapallana glaciers). Although local climatic conditions vary, approximately 2,330 glaciers in the Peruvian Andes have summit elevations of less than 5,500 m (as of 2014, calculated from ANA (2014)) and therefore face similar outlooks.

Lacking historical humidity and solar radiation records onsite, our temperature-index glacier module provides a reasonable estimate of glacier melt (Fernàndez & Mark, 2016) with the available temperature data and compares well ( $R^2 = 0.67$ ) with observed historical glacier areas by López-Moreno et al. (2014) (Figure 4a). However, the glacier module does not include edge-effect feedback, in which longwave radiation from rock surrounding the glacier plays an increasingly important role as the glacier retreats (Aubry-Wake et al., 2015), or elevation feedback, in which glacier thinning lowers the elevation of the glacier surface and is subjected to higher temperatures (Weertman, 1961). These unaccounted feedback make our glacier melt projections conservative estimates of mass loss through time.

#### 4.6. Groundwater Temporarily Buffers Loss of Glacier Melt Under Future Climatic Conditions

As glaciers recede, meltwater production initially increases to a maximum, known as "peak water," before dropping off as the glaciers continue to retreat (Gleick & Palaniappan, 2010). Approximately half of the world's glacierized drainage basins have passed peak water (Huss & Hock, 2018), including >70% of studied proglacial watersheds in the Cordillera Blanca, Peru (Baraer et al., 2012). Our model simulates multiple peaks in meltwater production, the last of which occurs when glacier areal retreat is fastest, around 2013, and then steadily drops off until it becomes zero when the glacier completely disappears (Figure 4b).

While the projected decrease in glacier melt in turn decreases stream discharge, it is partially mitigated by groundwater discharge to the stream. During the dry season, glacier melt steadily decreases after peak water. Groundwater discharge to the stream, on the other hand, remains relatively constant (Figure 4b), thereby maintaining a buffer to stream flow during the water-stressed dry season. However, as temperature increases, so does ET, which reduces groundwater recharge. Furthermore, changes in the frequency of extreme events and rain-snow fractions are implicitly included in the climate projections and also affect the amount of groundwater recharge. The effect of decreasing recharge is observed in the gradual decreasing trend in dry season groundwater discharge (Figure 4b). This effect is more severe in the RCP 8.5 scenario where temperature continues to increase through 2099, and where there is a severe reduction in precipitation for some GCMs.

Although both RCP 4.5 and 8.5 predict complete loss of glacier meltwater by the end of the twenty-first century, there is a marked difference in streamflow between the scenarios. RCP 4.5 (RCP 8.5) projects 20% (50%) lower streamflow and 18% (35%) lower groundwater discharge year round due to both lower precipitation and higher ET (Table S4 and Figure 4). Both future scenarios project that a larger proportion of streamflow will come from groundwater. These projections do not account for changes in ET as vegetation colonizes the



**Figure 4.** Climate, glacier, and hydrologic projections. (a) Projected change in maximum, minimum temperature, and glacier area from 1960 to 2099 and observed glacier areas reprocessed from López-Moreno (2014). (b) Modeled dry season streamflow, groundwater discharge, and glacier melt for the entire model domain (outlet at location 7 in Figure 1a). Peak glacier runoff occurs in 2013. (c) Change in modeled mean monthly streamflow between the first and last decade of the twenty-first century. (d) Change in groundwater discharge between the first and last decade of the twenty-first century. (e) The proportion of streamflow that comes from groundwater discharge, where a value of 1 means that all streamflow is sourced from groundwater discharge, demonstrating the seasonality of groundwater importance annually.

deglaciating area. However, the additional ET from the  $\sim 2\%$  glacierized area of the watershed is assumed to be relatively minor considering the cold, steep, and rocky environment.

All of the modeled climate scenarios project a decrease in water table elevation and groundwater storage (Table S5). On average, the groundwater table drops by 8.8 and 13.4 m for RCP 4.5 and 8.5, respectively, over the twenty-first century. The greatest changes in water table elevation occur beneath the Huaytapallana glaciers, where the loss of meltwater decreases groundwater recharge, and beneath the steepest hillslopes,

where hydraulic gradient is greatest. The least amount of change occurs near the stream network and in the high flat areas of the watershed. Similar spatial trends in groundwater table sensitivity to recharge have been suggested in the Alps (Ofterdinger et al., 2014).

#### 5. Conclusions

Though historically considered a minor component of mountain streamflow (Liu et al., 2004), our results demonstrate the current and future importance of groundwater in a mountain hydrologic system. Groundwater dominates (72%) dry season streamflow, including substantial contribution from fractured bedrock aquifers, but receives little recharge from glacier melt (2%), making groundwater storage resilient to decreases in glacier melt. In the near term (~30 years), consistent groundwater discharge mediates the loss of glacier melt. But, in the long term (>60 years), increasing ET and decreasing precipitation (for some GCMs) decrease groundwater recharge and therefore stream discharge, particularly for the high carbon emission scenarios (Figure 1). The relatively small (~0–2 m) projected drop in water table beneath flat areas may lend some resilience to high-altitude wetlands located in these areas. However, Andean wetland ecosystems, considered biodiversity hot spots, may be quite sensitive to changes in water table (Maldonado Fonkén, 2014).

More broadly, our results demonstrate that mountain water resource response to climate change is shaped not only by glacier change but also by surface and—particularly—groundwater hydrological processes. These findings provide important context to large-scale glaciological studies (e.g., Huss & Hock, 2018). Future efforts to model mountain hydrology under climate change should include a realistic groundwater component.

#### Acknowledgments

The authors thank SENAMHI, ANA, and IGP for the collection of meteorological and hydrological data and Sin Chan Chou for the support in acquiring regional climate data from the CPTEC repository. The National Science Foundation and the United States Agency for International Development provided funding as part of the NSF-USAID PEER Program (project 3-127), NSF EAR-1316432, and additional funding was provided by the Natural Science and Engineering Research Council of Canada. We would also like to thank Richard Niswonger for the technical advice in implementing GSFLOW and acknowledge the help of our field assistants: Américo and Alvaro González Caldua, Oliver Wigmore, Pierrick Lamontagne-Hallé, Nadine Shatilla, and Emilio Mateo. Input and output files for GSFLOW, Matlab code for the glacier melt module, and observed streamflow, groundwater, and glacier data are available from our online repository: https:// zenodo.org/record/3465975#. XZJOylVKi70.

#### References

- Andermann, C., Longuevergne, L., Bonnet, S., Crave, A., Davy, P., & Gloaguen, R. (2012). Impact of transient groundwater storage on the discharge of Himalayan rivers. *Nature Geoscience*, 5, 2012. https://doi.org/10.1038/NGEO1356
- Aubry-Wake, C., Baraer, M., McKenzie, J. M., Mark, B. G., Wigmore, O., Hellström, R. A., & Lautz, L. (2015). Measuring glaciersurface temperatures with ground-basedthermal infrared imaging. *Geophysical Research Letters*, 42, 8489–8497. https://doi.org/10.1002/ 2015GL065321
- Authoridad Nacional del Agua (2014). Inventario de glaciares del Peru. Huaraz, Peru: Ministerio de agricultura y riego.
  - Bales, R. C., Molotch, N. P., Painter, T. H., Dettinger, M. D., Rice, R., & Dozier, J. (2006). Mountain hydrology of the western united states. Water Resources Research, 42, W08432. https://doi.org/10.1029/2005WR004387
  - Baraer, M., Mark, B. G., McKenzie, J. M., Comdom, T., Bury, J., Huh, K., et al. (2012). Glacier recession and water resources in Peru's Cordillera Blanca. *Journal of Glaciology*, 58(207), 134–149.
  - Baraer, M., McKenzie, J., Mark, B. G., Gordon, R., Bury, J., Condom, T., et al. (2015). Contribution of groundwater to the outflow from ungauged glazierized catchments: a multi-site study in the tropical Cordillera Blanca, Peru. Hydrological Processes, 29(11), 2561–2581.
- Barnett, T. P., Adam, J. C., & Lettenmaier, D. P. (2005). Potential impacts of a warming climate on water availability in snow-dominated regions. *Nature*, 438(7066), 303–309. https://doi.org/10.1038/nature04141
- Bordoy, R., & Burlando, P. (2013). Bias Correction of Regional Climate Model Simulations in a Region of Complex Orography. Journal of Applied Meteorology and Climatology, 52, 82–101. https://doi.org/10.1175/JAMC-D-11-0149.1
- Bradley, R. S., Vuille, M., Diaz, H. F., & Vergara, W. (2006). Threats to water supplies in the tropical Andes. *Science*, 312(5781), 1755–1756. https://doi.org/10.1126/science.1128087
- Bury, J., Mark, B. G., Carey, M., Young, K. R., McKenzie, J. M., Baraer, M., et al. (2013). New geographies of water and climate change in Peru: Coupled natural and social transformations in the Santa river watershed. *Annals of the Association of American Geographers*, 103 (2), 363–374.
- Chew, D. M., Pedemonte, G., & Corbett, E. (2016). Proto-Andea evolution of the Eastern Cordillera of Peru. Gondwana Research, 35(2016), 59–78. http://doi.org/10.1016/j.gr.2016.03.016
- Chou, S. C., Lyra, A., Mourão, C., Dereczynski, C., Pilotto, I., Gomes, J., et al. (2014). Assessment of Climate Change over South America under RCP 4.5 and 8.5 Downscaling Scenarios. *American Journal of Climate Change*, 3, 512–527. https://doi.org/10.4236/ ajcc.2014.35043
- Chou, S. C., Lyra, A., Mourão, C., Dereczynski, C., Pilotto, I., Gomes, J., et al. (2014). Evaluation of the Eta Simulations Nested in Three Global Climate Models. *American Journal of Climate Change*, *3*, 438–454. https://doi.org/10.4236/ajcc.2014.35039
- Crumley, R. (2015). Investigating glacier melt contribution to stream discharge and experiences of climate change in the Shullcas River Watershed in Peru. Masters Thesis, The Ohio State University, Columbus, USA.
- Cuthbert, M. O., Gleeson, T., Moosdorf, N., Befus, K. M., Schneider, A., Hartmann, J., & Lehner, B. (2019). Global patterns and dynamics of climate-groundwater interactions. *Nature Climate Change*, 9(2), 137–141. http://Doi.org/10.1038/s41558-018-0386-4
- Fernàndez, A., & Mark, B. G. (2016). Modeling modern glacier response to climate changes along the Andes Cordillera: A multi-scale review. *Journal of Advances in Modeling Earth Systems*, *8*, 467–495. https://doi.org/10.1002/2015MS000482
- Frisbee, M. B., Phillips, F. M., Campbell, A. R., Liu, F., & Sanchez, S. A. (2011). Streamflow generation in a large, alpine watershed in the southern Rocky Mountains of Colorado: Is streamflow generation simply the aggregation of hillslope runoff responses? Water Resources Research, 47, W06512. http://doi.org/10.1029/2010WR009391
- Glas, R., Lautz, L., McKenzie, J., Mark, B., Baraer, M., Chavez, D., & Maharaj, L. (2018). A review of the current state of knowledge of proglacial hydrogeology in the Cordillera Blanca, Peru. WIREs Water, 5, c1299. https://doi.org/10.1002/wat2.1299
- Gleick, P. H., & Palaniappan, M. (2010). Peak water limits to freshwater withdrawal and use. Proceedings of the National Academy of Sciences of the United States of America, 107(25), 11155–11162. https://doi.org/10.1073/pnas.1004812107



- Gordon, R. P., Lautz, L. K., McKenzie, J. M., Mark, B. G., Chavez, D., & Baraer, M. (2015). Sources and pathways of stream generation in tropical proglacial valleys of the Cordillera Blanca, Peru. *Journal of Hydrology*, 522, 628–644. https://doi.org/10.1016/j. jhydrol.2015.01.013
- Grillakis, M. G., Koutroulis, A. G., & Tsanis, I. K. (2013). Multisegment statistical bias correction of daily GCM precipitation output. Journal of Geophysical Research: Atmospheres, 118, 3150–3162. https://doi.org/10.1002/jgrd.50323
- Hawkins, E., Osborne, T. M., Ho, C. K., & Challinor, A. J. (2012). Calibration and bias correction of climate projections for crop modelling: An idealised case study over Europe. Agricultural and Forest Meteorology, 170, 19–31. https://doi.org/10.1016/j.agrformet.2012.04.007
- Hock, R. (2003). Temperature index melt modelling in mountain areas. *Journal of Hydrology*, 282, 104–115. https://doi.org/10.1016/S0022-1694(03)00257-9
- Hood, J., & Hayashi, M. (2015). Characterization of snowmelt flux and groundwater storage in an alpine headwater basin. Journal of Hydrology, 521(2015), 482–497. https://doi.org/10.1016/j.jhydrol.2014.12.041
- Huffman, G. J., & Bolvin, D. T. (2015). TRMM and other data precipitation data set documentation. National Aeronautics and Space Administration (NASA): Greenbelt USA.
- Huss, M., Farinotti, D., Bauder, A., & Funk, M. (2008). Modelling runoff from highly glacierized alpine drainage basins in a changing climate. *Hydrological Processes*, 22(19), 3888–3902. https://doi.org/10.1002/hyp.7055
- Huss, M., & Hock, R. (2018). Global-scale hydrological response to future glacier mass loss. *Nature Climate Change*, 8, 135–140. https://doi.org/10.1038/s41558-017-0049-x
- Ines, A. V. M., & Hansen, J. W. (2006). Bias correction of daily GCM rainfall for crop simulation studies. Agricultural and Forest Meteorology, 138, 44–53.
- Instituto Nacional de Estadisticas y Informaticas Perú, (2012). Estiamciones y Proyecciones de Población por sexo, según Departamento, Provincia y Distrito, 2000 – 2015, Lima, Peru.
- Juen, I., Kaser, G., & Georges, C. (2007). Modelling observed and future runoff from a glacierized tropical catchment (Cordillera Blanca, Peru). *Global and Planetary Change*, 59, 37–48. http://Doi.org/10.1016/j.gloplacha.2006.11.038
- Kaser, G., Cogley, J. G., Dyurgerov, M. B., Meier, M. F., & Ohmura, A. (2006). Mass balance of glacier and ice caps: Consensus estimates for 1961-2004. Geophysical Research Letters, 33, L19501. http://Doi.org/10.1029/2006GL027511
- Langevin, C. D., Hughes, J. D., Banta, E. R., Niswonger, R. G., Panday, Sorab, & Provost, A. M. (2017). Documentation for the MODFLOW 6 Groundwater Flow Model: U.S. Geological Survey Techniques and Methods, book 6 (Chap. A55, pp. 197). Reston Virginia, USA: USGS. https://doi.org/10.3133/tm6A55
- Lemieux, J. M., Sudicky, E. A., Peltier, W. R., & Tarasov, L. (2008). Simulating the impact of glaciations on continental groundwater flow systems: 1. Relevant processes and model formulation. *Journal of Geophysical Research*, 113, F03017. https://doi.org/10.1029/ 2007JF000928
- Liu, F., Williams, M. W., & Caine, N. (2004). Source waters and flow paths in an alpine catchment, Colorado Front Range, United States. Water Resources Research, 40, W09401. https://doi.org/10.1029/2004WR003076
- Liu, X., Xu, C., Zhong, X., Li, Y., Yuan, X., & Cao, J. (2017). Comparison of 16 models for reference crop evapotranspiration against weighing lysimeter measurement. Agricultural Water Management, 184, 145–155. https://doi.org/10.1016/j.agwat.2017.01.017
- López-Moreno, J. I., Fontaneda, S., Bazo, J., Revuelto, J., Azorin-Molina, C., Valero-Garcés, B., et al. (2014). Recent glacier retreat and climate trends in Cordillera Huaytapallana, Peru. *Global and Planetary Change*, *112*, 1–11.
- Maldonado Fonkén, M. S. (2014). An introduction to the bofedales of the Peruvian High Andes. Mires and Peat, 15(5), 1–13.
- Mark, B. G., French, A., Baraer, M., Carey, M., Bury, J., Young, K. R., et al. (2017). Glacier loss and hydro-social risks in the Peruvian Andes. Global and Planetary Change, 159, 61–76.
- Markovich, K. H., Maxwell, R. M., & Fogg, G. E. (2016). Hydrogeological response to climate change in alpine hillslopes. *Hydrological Processes*, *30*, 3126–3138. http://doi.org/10.1002/hyp.10851
- Markstrom, S.L., Niswonger, R.G., Regan, R.S., Prudic, D.E. & Barlow, P.M. (2008). GSFLOW-Coupled Ground-Water and Surface-Water Flow Model Based on the Integration of the Precipitation-Runoff Modeling System (PRMS) and the Modular Ground-Water Flow Model (MODFLOW-2005). U.S. Geological Survey Techniques and Methods 6-D1, .240.
- Markstrom, S.L., Regan, R. S., Hay, L. E., Viger, R. J., Webb, R. M. T., Payn, R. A., & LaFontaine, J. H. (2015). PRMS-IV, the precipitation-runoff modeling system, version 4: U.S. Geological Survey Techniques and Methods, book 6 (Chap. B7,158). Reston, Virginia, USA: USGS. http://Doi.org/10.3133/tm6B7
- Mather, P. M. (1985). A computationally-efficient maximum-likelihood classifier employing prior probabilities for remotely-sensed data. International Journal of Remote Sensing, 6(2), 369–376.
- McClymont, A. F., Hayashi, M., Bentley, L. R., Muir, D., & Ernst, E. (2010). Groundwater flow and storage within an alpine meadow-talus complex. *Hydrology and Earth System Sciences*, 14(6), 859–872. https://doi.org/10.5194/hess-14-859-2010
- Moriasi, D. N., Arnold, J. G., Van Liew, M. W., Bingner, R. L., Harmel, R. D., & Veith, T. L. (2007). Model evaluation guidelines for systematic quantification of accuracy in watershed simulations. *American Society of Agricultural and Biological Engineers*, 50(3), 885–900. Mountain Research Initiative EDW Working Group (2015). Elevation-dependent warming in mountain regions of the world. *Nature*
- Climate Change, 5, 424–430. https://doi.org/10.1038/NGEO1356
- Ng, G.-H. C., Wickert, A. D., Somers, L. D., Saberi, L., Cronkite-Ratcliff, C., Niswonger, R. G., & McKenzie, J. M. (2018). GSFLOW-GRASS v1.0.0: GIS-enabled hydrologic modeling of coupled groundwater–surface-water systems. *Geoscience Model Development*, *11*, 4755–4777. https://doi.org/10.5194/gmd-11-4755-2018
- Ofterdinger, U. S., Renard, P., & Loew, S. (2014). Hydraulic subsurface measurements and hydrodynamic modelling as indicators for groundwater flow systems in the Rotondo granite, Central Alps (Switzerland). *Hydrological Processes*, *28*, 255–278. https://doi.org/ 10.1002/hyp.9568
- Paul, F., Kääb, A., Maisch, M., Kellenberger, T., & Haeberli, W. (2004). Rapid disintegration of Alpine glaciers observed with satellite data. Geophysical Research Letters, 31, L21402. https://doi.org/10.1029/2004GL020816
- Pohl, E., Knoche, M., Gloaguen, R., Andermann, C., & Krause, P. (2015). Sensitivity analysis and implications for surface processes from a hydrological modelling approach in the Gunt catchment, high Pamir Mountains. *Earth Surface Dynamics*, 3, 333–362. http://Doi.org/ 10.5194/esurf-3-333-2015
- Priestley, C. H. B., & Taylor, R. J. (1972). On the assessment of surface heat flux and evaporation using large-scale parameters. *Monthly Weather Review*, 100(2), 81–92.
- Quick, M. C., & Pipes, A. (1977). UBC watershed model. Hydrological Sciences Bulletin, 221, 153-161.
- Rabatel, A., Francou, B., Soruco, A., Gomez, J., Cáceres, B., Ceballos, J. L., et al. (2013). Current state of glaciers in the tropical Andes: A multi-century perspective on glacier evolution and climate change. *The Cryosphere*, 7(1), 81–102.

- Ragettli, S., Cortés, G., McPhee, J., & Pellicciotti, F. (2014). An evaluation of approaches for modelling hydrological processes in highelevation, glacierized Andean watersheds. *Hydrological Processes*, 28(2014), 5674–5695. https://doi.org/10.1002/hyp.10055
- Saberi, L., McLaughlin, R., Ng, G.-H. C., La Frenierre, J., Wickert, A., Baraer, M., et al. (2019). Multi-scale temporal variability in meltwater contributions in a tropical glacierized watershed. *Hydrology and Earth System Sciences*, 23, 405–425. https://doi.org/10.5194/hess-23-405-2019
- Salvador, F., Monerris, J., & Rochefort, L. (2014). Pealands of the Peruvian Puna ecoregion: Types, characteristics and disturbance. *Mires and Peat*, 15(3), 1–17.
- SENAMHI (2018). Datos Hidrometeorológicos. Retrieved from https://www.senamhi.gob.pe/?&p=estaciones, Sept., 2018.
- Seth, A., Thibeault, J., Garcia, M., & Valdivia, C. (2010). Making sense of twenty-first-century climate change in the Altiplano: Observed trends in CMIP3 projections. Annals of the Association of American Geographers, 100(4), 835–847.
- Smitha, P. S., Narasimhan, B., Sudheer, K. P., & Annmalai, H. (2018). An improved bias correction method of daily rainfall data using a sliding window technique for climate change impact assessment. *Journal of Hydrology*, 556, 100–118. https://doi.org/10.1016/j. jhydrol.2017.11.010
- Somers, L., Gordon, R. P., Lautz, L. K., McKenzie, J. M., Wigmore, O., Glose, A., et al. (2016). Quantifying groundwater-surface water interactions in a proglacial valley, Cordillera Blanca, Peru. *Hydrological Processes*. https://doi.org/10.1002/hyp.10912
- Tachikawa, T., Kaku, M., Iwasaki, A., Gesch, D., Oimoen, M., Zhang, Z. & Meyer, D. (2011). ASTER Global Digital Elevation Model version 2 —Summary of validation results. NASA Land Processes Distributed Active Archive Center and the Joint Japan-US ASTER Science Team. (https://lpdaacaster.cr.usgs.gov/GDEM/Summary\_GDEM2\_validation\_report\_final.pdf)
- Tague, C., Grant, C., Farrell, M., Choate, J., & Jefferson, A. (2008). Deep groundwater mediates streamflow response to climate warming in the Oregon Cascades. *Climatic Change*, *86*, 189–210.
- United Nations Development Report (UNDP) (2006). Human Development Report, beyond scarcity: Power, poverty and the global water crisis. New York, NY, USA: Palgrave MacMillan. ISBN: 0-230-50058-7.
- Urrutia, R., & Vuille, M. (2009). Climate Change projections for the tropical Andes using a regional climate model: Temperature and precipitation simulations for the end of the 21st century. *Journal of Geophysical Research*, 114, D02108. https://doi.org/10.1029/ 2008JD011021
- Voeckler, H. M., Alen, D. M., & Alila, Y. (2014). Modeling coupled surface water Groundwater processes in a small mountainous headwater catchment. *Journal of Hydrology*, 517, 1089–1106. https://doi.org/10.1016/j.jhydrol.2014.06.015
- Vuille, M., Carey, M., Huggel, C., Buytaert, W., Rabatel, A., Jacobsen, D., et al. (2018). Rapid decline of snow and ice in the tropical Andes Impacts, uncertainties and challenges ahead. *Earth-Science Reviews*, 176, 195–213.
- Viviroli, D., Dürr, H. H., Messerli, B., Meybeck, M., & Weingartner, R. (2007). Mountains of the world, water towers for humanity: Typology, mapping, and global significance. *Water Resources Research*, 43(7), W07447. https://doi.org/10.1029/2006WR005653
- Weertman, J. (1961). Stability of ice-age ice sheets. Journal of Geophysical Research, 66, 3783–3792. https://doi.org/10.1029/ JZ066i011p03783
- Yarleque, C., Vuille, M., Hardy, D. R., Timm, O. E., De la Cruz, J., Ramos, H., & Rabatel, A. (2018). Projections of the future disappearance of the Quelccaya Ice Cap in the Central Andes. *Scientific Reports*, 8(1), 1–11. http://doi.org/10.1038/s41598-018-33698-z
- Zheng, C., & Bennett, G. D. (2002). Applied Contaminant Transport Modeling, (2 nd ed.pp. 343–347). New York, NY: John Wiley and Sons Inc.

#### **References From the Supporting Information**

- Eaton, A., Rouse, W. R., Lafleur, P. M., Marsh, P., & Blanken, P. D. (2001). Surface energy balance of the western and central Canadian subarctic: variations in the energy balance among five major terrain types. *Journal of Climate*, 14, 3692–3703.
- Frisbee, M. D., Tolley, D. G., & Wilson, J. L. (2017). Field estimates of groundwater circulation depths in two mountainous watersheds in the western U.S. and the effect of deep circulation on solute concentrations in streamflow. *Water Resources Research*, 53, 2693–2715. https://doi.org/10.1002/2016WR019553
- Hamimed, A., Benslimane, M., Khaldi, A., Nehal, L., & Zaagane, M. (2014). Using the Priestley-Taylor expression for estimating actual evapotranspiration and monitoring water stress from satellite landsat ETM+ data. *Larhyss Journal*, *18*, 179–195.
- Heath, R.C. (1983). Basic ground-water hydrology, U.S. Geological Survey Water-Supply Paper 2220, 86p.
- Kaser, G., & Osmaston, H. (2002). Tropical Glaciers. Cambridge, UK: Cambridge University Press.
- Saunders, I. R., Bailey, W. G., & Bowers, J. D. (1997). Evaporation regimes and evaporation modelling in an alpine tundra environment. Journal of Hydrology, 195, 99–113.
- Teng, J., Vaze, J., Chiew, F. H., Wang, B., & Perraud, J. (2012). Estimating the Relative Uncertainties Sourced from GCMs and Hydrological Models in Modeling Climate Change Impact on Runoff. Journal of Hydrometeorology, 13, 122–139. https://doi.org/10.1175/JHM-D-11-058.1